### **DISCUSSION PAPER SERIES**

No. 9190

#### THE BRAIN GAIN OF CORPORATE BOARDS: A NATURAL EXPERIMENT FROM CHINA

Mariassunta Giannetti, Guanmin Liao and Xiaoyun Yu

DEVELOPMENT ECONOMICS and FINANCIAL ECONOMICS



### Centre for Economic Policy Research

### www.cepr.org

Available online at:

www.cepr.org/pubs/dps/DP9190.asp

### THE BRAIN GAIN OF CORPORATE BOARDS: A NATURAL EXPERIMENT FROM CHINA

#### Mariassunta Giannetti, Stockholm School of Economics and CEPR Guanmin Liao, Central University of Finance and Economics Xiaoyun Yu, Indiana University

Discussion Paper No. 9190 October 2012

Centre for Economic Policy Research 77 Bastwick Street, London EC1V 3PZ, UK Tel: (44 20) 7183 8801, Fax: (44 20) 7183 8820 Email: cepr@cepr.org, Website: www.cepr.org

This Discussion Paper is issued under the auspices of the Centre's research programme in **DEVELOPMENT ECONOMICS and FINANCIAL ECONOMICS**. Any opinions expressed here are those of the author(s) and not those of the Centre for Economic Policy Research. Research disseminated by CEPR may include views on policy, but the Centre itself takes no institutional policy positions.

The Centre for Economic Policy Research was established in 1983 as an educational charity, to promote independent analysis and public discussion of open economies and the relations among them. It is pluralist and non-partisan, bringing economic research to bear on the analysis of medium- and long-run policy questions.

These Discussion Papers often represent preliminary or incomplete work, circulated to encourage discussion and comment. Citation and use of such a paper should take account of its provisional character.

Copyright: Mariassunta Giannetti, Guanmin Liao and Xiaoyun Yu

CEPR Discussion Paper No. 9190

October 2012

### ABSTRACT

# The Brain Gain of Corporate Boards: a Natural Experiment from China\*

We study the impact of directors with foreign experience on firms in emerging markets. To establish causality, we use a unique dataset from China and exploit that at different times, Chinese provinces introduced policies to attract highly talented emigrants. These policies led to an exogenous increase in the supply of Chinese individuals with foreign experience in the local labor market and ultimately increased the likelihood that firms in these provinces had directors with foreign experience in comparison to firms with a similarly high demand for these skills elsewhere. We document that hiring directors with foreign experience results in higher firm valuation, productivity, and profitability. Furthermore, corporate governance improves and firms are more likely to make international acquisitions, to export, and to raise funds internationally. These results indicate that the transfer of knowledge to emerging markets occurs not only through foreign investment, but also through labor flows and, in particular, return migration.

JEL Classification: D22, D80, F21, F22, G30 and J24 Keywords: corporate boards, corporate governance, firm performance, firm productivity and human capital

Mariassunta Giannetti Department of Finance Stockholm School of Economics Sveavägen 65, Box 6501 SE-113 83 Stockholm SWEDEN	Guanmin Liao Central University of Finance and Economics 39 Xueyuan South Road Haidian, Beijing
Email: mariassunta.giannetti@hhs.se	Email: liaoguanmin@cufe.edu.cn
For further Discussion Papers by this author see: www.cepr.org/pubs/new-dps/dplist.asp?authorid=147797	For further Discussion Papers by this author see: www.cepr.org/pubs/new-dps/dplist.asp?authorid=176394

Xiaoyun Yu Kelly School of Business Indiana University 1309 East Tenth Street Bloomington, IN 47405-1701 USA

Email: xiyu@indiana.edu

For further Discussion Papers by this author see: www.cepr.org/pubs/new-dps/dplist.asp?authorid=155203

\*We thank Laurent Bach, Walid Busaba, Chun Chang, Jess Cornaggia, Yaniv Grinstein, Juanna Joensen, Daniel Metzger, Wei-Ling Song, Frank Yu, and seminar participants at the Western Finance Association annual meeting (Las Vegas), the European Finance Association annual meeting (Copenhagen), EDHEC, ESSEC, Fudan University, Indiana University, Louisiana State University, Shanghai Advanced Institute of Finance at Shanghai Jiaotong University, Vienna Graduate School of Finance, Stockholm School of Economics (SITE), University of St. Gallen, University of Nottingham, and University of Western Ontario for helpful comments. Giannetti acknowledges financial support from the Jan Wallander and Tom Hedelius Foundation and the Bank of Sweden Tercentenary Foundation. Liao acknowledges financial support from the National Natural Science Foundation of China, Grant # 70902001. This paper was started when Liao was visiting Indiana University, which we thank for the generous hospitality. Yu acknowledges financial support from CIBER at Indiana University.

Submitted 16 October 2012

#### **1. INTRODUCTION**

Development economists have long warned about the costs for developing countries of the emigration of the best and brightest that decamp to universities and businesses in the developed world (Bhagwati, 1976). While this *brain drain* has attracted a considerable amount of economic research, more recently, arguments have been raised that the emigration of the brightest may actually benefit developing countries, because emigrants may eventually return with more knowledge and organizational skills (Docquier and Rapoport, 2012; Nanda and Khanna, 2010).<sup>1</sup> Thus, the brain drain may actually become a *brain gain*.

In this paper, we highlight a specific channel through which the brain gain arising from return migration to emerging markets may benefit the overall economy: the brain gain in the corporate boards of publicly listed companies. Specifically, we explore the effects of individuals with foreign experience joining the boards of directors on firms' performance and corporate policies in China.

We expect that the arrival of individuals with foreign experience in the board of listed companies in emerging markets can enhance the firms' productivity and performance, because board members perform an important advisory role for the management (Fama and Jensen, 1983). Board members with foreign experience, having learnt how foreign organizations work, may facilitate the adoption of superior management practices, which—as shown by Bloom and Van Reenen (2007)—greatly enhance the performance and productivity of firms. Board directors with foreign experience could thus help bridging the large productivity gaps that persist across countries and firms (Hall and Jones, 1999; Jones and Romer, 2009).

Directors with foreign experience may also perform more effectively the monitoring function of the board and help improve firm level corporate governance in emerging markets,

<sup>&</sup>lt;sup>1</sup> See The Economist, May 26, 2011.

not only thanks to the foreign expertise accumulated abroad, but also because, being relatively disenfranchised from local ties, they may have stronger incentives to pursue profitability rather than pleasing politicians and other local constituencies.

It is empirically challenging to establish whether directors with foreign experience have a causal effect on firm performance. In particular, firms may select or attract directors with foreign experience because they have certain characteristics (e.g., growth opportunities) that independently affect firm performance. We surmount this challenge by using a unique dataset from China and by exploiting exogenous variation caused by policy changes. We handcollect information on foreign education, work experience and other demographic characteristics from the bios of 33,707 executive and non-executive directors of 1,733 publicly listed companies from 1999 to 2009. We consider an individual to have foreign experience if he or she studied and/or worked outside (mainland) China.

China provides a unique environment to address the endogeneity problems for the following reasons. First, Chinese firms face talent shortage in filling managerial positions. Since individuals with foreign experience are scarce, not all firms with similarly high demand for directors with foreign experience are able to attract one. Second and most importantly, during the sample period, almost all provinces, at different times, introduced incentives for highly skilled individuals with foreign experience to return. Since the labor market for board directors is largely local (Knyazeva, Knyazeva and Masulis, 2011), the introduction of the provincial policies determined exogenous changes in the supply of potential directors with foreign experience.

Not being directed at listed companies, the timing of the introduction of the incentives was largely independent from the characteristics and growth opportunities of the publicly listed firms in our sample (an assumption that we test in a number of ways described below). Instead, it was determined by how politically progressive the provincial government was. We show that after the policy changes, the number of directors with foreign experience increases more for the firms headquartered in the provinces adopting the policies than for comparable firms elsewhere. This is the case not only because some individuals return and become executive directors of the company, but mostly because there is a larger pool of individuals with foreign experience working in the area, who can become independent directors. The increase in board members with foreign experience provoked by the provincial policy changes is extremely unlikely to have coincided with changes in the demand for directors with these skills (or with other firm-level changes) for firms affected by the policies in comparison to similar firms headquartered elsewhere. We can thus conclude that directors with foreign experience *enable* the changes in corporate policies concurrent to their arrival in the board of a firm.

We find that when individuals with foreign experience join the board of a company, the firm's valuation improves and its total factor productivity increases. Furthermore, in the subsequent years, the firm's profitability increases. We also show that these improvements in performance are accompanied by changes in corporate policies that are generally set by the board. First, firms' propensity to manage earnings decreases and CEO turnover following low profits increases, indicating that corporate governance improves. Second, among the firms that make mergers and acquisitions, the ones with board members with foreign experience are more likely to make an international merger or acquisition. This suggests that these firms are able to access a broader range of investment opportunities. Similarly, firms with board members with foreign experience are able to access more sources of external financing, as they are more likely to engage a foreign investor when raising capital through private placements than other firms without directors with foreign experience. Lastly, firms that hire directors with foreign experience start exporting more. Overall, these results suggest that firm performance improves because, among other effects, directors with foreign

experience facilitate the adoption of strong corporate governance practices and internationalization.

We provide a number of further tests supporting the mechanism behind the causal interpretation of the empirical evidence. First, we estimate alternative models including province fixed effects, firm fixed effects, and controls for firm previous performance and changing economic conditions across provinces and across industries over time. All these tests, even when we exploit only differences in the firms' ability to attract directors with foreign experience within the same province for the identification, are fully consistent with our previous results and strongly support the positive causal effect of board members with foreign experience on firm performance.

Second, we show that the policy changes affect positively firm performance only for firms that actually end up hiring a director with foreign experience, thus supporting our identifying assumption that the increase in supply is truly exogenous and unrelated to province-specific growth opportunities. This evidence also suggests that hiring directors with foreign experience is crucial and that our results are unlikely to be driven by similar talent hired at a lower rank.

Finally, since individuals with foreign experience are a relatively scarce resource in China and there exist companies that would like to employ directors with foreign experience, but are unable to do so for idiosyncratic reasons, we can use propensity scores and compare the performance of similar companies with and without directors with foreign experience. In other tests, we restrict the sample to firms that employ at least one director with foreign experience during the sample period. All these tests, in which we restrict the control sample to firms that are more homogeneous and more likely to experience the same shocks as the firms that actually hire directors with foreign experience, consistently indicate a positive impact of directors with foreign experience on firm performance. This confirms that these directors are crucial to enable the changes in corporate policies.

This paper is related to a growing literature exploring the effects of board expertise and structure on performance (e.g., Cole, Daniel and Naveen, 2008; Klein, 1998). Adams, Hermalin and Weisbach (2010) provide a recent and comprehensive survey of this literature. Ahearn and Dittmar (2011) explore the effect of gender quotas in Norway on changes in board composition and firm performance. Particularly related to ours are papers exploring how directors' expertise affects corporate decisions and corporate governance. Most of these papers, spurred by the Sarbanes Oxley Act, focus on the effects of board independence and financial expertise (Agrawal and Chadha, 2005; Chhaochharia and Grinstein, 2009; Güner, Malmendier and Tate, 2008; Guthrie, Sokolowsky and Wan, 2012). To the best of our knowledge, this is the first paper exploring the effect of foreign experience and returnee migrants in corporate boards. Furthermore, while most of the existing literature focuses on the US, we explore the role of boards in an emerging market. Since underdeveloped financial systems lack sophisticated gatekeepers (e.g., domestic and foreign institutional investors and analysts) monitoring listed companies (Stulz, 1999), the impact of the board is potentially larger. Furthermore, directors could transmit know-how and managerial skills that are known to improve firm performance in emerging markets (Bloom et al., 2012).

Our work is also related to the literature exploring how foreign investment and foreign acquisitions improve firm performance (e.g., Guadalupe, Kuzmina and Thomas, 2012). We highlight that these benefits can accrue not only through foreign investment, but also by engaging individuals with certain skills in firm management and monitoring.

The rest of the paper is organized as follows. Section 2 discusses the institutional background in China and our research setting. Section 3 describes our identification strategy. Section 4 introduces our data sources and sample construction. Sections 5 and 6 present the

empirical results. Section 7 discusses the robustness tests. Section 8 concludes the paper. Variable definitions are in the Appendix.

#### 2. BACKGROUND

#### 2.1. The Chinese Environment

China is the largest emerging market and has experienced spectacular economic growth since the late 1970s, when it initiated an overhaul of its economic system. While the economy has a large surplus of unskilled labor, there is significant talent shortage. Multinational companies and domestic firms find that few Chinese university graduates have the necessary skills for service occupations, such as engineers, finance workers, accountants, quantitative analysts, generalists and life science researchers (Farrell and Grant, 2007). Little practical experience in projects or teamwork, poor English and, more in general, poor communication style and cultural fits are commonly adduced as limitations of local hires.

Farrell and Grant estimate that over the next 10 to 15 years, firms active in China will need 75,000 managers who can work effectively in an international environment; however, today they have only 3,000 to 5,000, mostly consisting of highly skilled returnee emigrants, who have worked or studied in developed economies.

Another problem constraining the growth of Chinese firms is poor corporate governance and, in particular, poor disclosure (Green, 2003; Gul, Kim and Qiu, 2010). In this regard, the board of directors may perform an important monitoring role. Newcomers that have been exposed to governance practices in developed countries may educate and coax the older guard of directors to begin to adhere to international standards of governance (Khanna, 2008).

Scarcity of managerial talent and poor corporate governance are problems common to many emerging markets. For this reason, we believe that from the experience of China, we can draw broader conclusions on the effects that directors with foreign experience, and more in general, labor flows and return migration have on firm performance and corporate governance.

#### 2.2. The Policies to Attract Highly Skilled Emigrants

While the Chinese political elite often had periods of studies abroad, the flows of students from China towards universities in the developed world became sizable in the early 1990s. After completing their studies, many Chinese immigrants also gained foreign work experience. Starting from the early 2000s, tens of thousands of individuals trained abroad have been returning to China. According to the China Statistical Yearbook 2006, while the number of individuals with foreign training returning to China in 1995 was about 5,000, the number of returnees had reached 35,000 in 2006. These returnees, often called "sea turtles", are mostly foreign-trained scientists and academics, who once in China may join corporate boards as dependent or independent directors.

The inversion of the brain drain was fostered by an environment of economic growth and political stability. However, government policies and inter-regional competition have favored the process. Starting in late 1990s, at different points in time, local governments adopted policies to attract highly skilled emigrants (Zweig, 2006). The policies adopted by the local governments include tax breaks, subsidized housing, tax-free imports of automobiles and computers, schooling for the children of the returnees, local grants and awards, medical benefits, jobs for spouses, and long-term residence permits. Only the most distinguished Chinese expatriates can benefit from these incentives.

The main objective of the policies was increasing the quality of academic and industrial research in China as well as fostering entrepreneurial activity and the entry of new businesses. In no case, the policies were targeted to publicly listed companies and their

7

boards. However, since directors in China are often university professors or researchers (Jiang, Wan and Zhao, 2011), the policies created a large pool of skilled individuals from which the board members of listed companies could be drawn.

We collect data on the timing of the incentives introduced by the local governments in mainland China from Wang, Zeng and Pu (2011), and verify the timing and type of policies through internet searches.<sup>2</sup> Table 1 provides detailed information on the timing of the policies. It is apparent that an earlier adoption of policies for reverting the brain drain is not necessarily related to regional economic development: While the highly developed Beijing and Guangdong were early adopters (in 2000 and 1999, respectively), so were the far less developed Inner Mongolia and Yunnan (in 2001). The highly developed Shanghai, on the other hand, implemented similar policies only in 2005.

Figure 1 provides a first glance at the relevance of the policies for the changes in board composition. Besides showing the fraction of board members with foreign experience for all provinces, before and after the policy adoption, it also shows the effect for Beijing, Shanghai and Guangdong, the provinces that attract most of the internal white and blue collar migration.<sup>3</sup> We observe a rise in the fraction of directors with foreign experience after the implementation of the provincial policies. On the aggregate, while in 1999 only 200 sample firms had at least one director with foreign experience, by the end of the sample 803 firms had at least one director with foreign experience (not tabulated).

#### **3. IDENTIFICATION STRATEGY**

Identifying the causal effect of board expertise on firm performance poses serious challenges because firms choose board structure optimally. The proportion of directors with

<sup>&</sup>lt;sup>2</sup> Our sample includes firms located in 22 provinces, 5 autonomous regions, and 4 municipalities (Beijing, Chongqing, Tianjin, and Shanghai). The municipalities are directly governed by the central government and enjoy the same status as provinces and autonomous regions.

<sup>&</sup>lt;sup>3</sup> The results we report hereafter are not driven exclusively by Beijing, Shanghai and Guangdong as the estimates are invariant if we exclude all firms from these three regions from the sample.

foreign experience depends on firm characteristics. In particular, firms that are in the process of implementing certain changes and that would grow more, irrespectively from their board composition, could select or attract board members with foreign experience.

We argue that our sample of Chinese firms is ideal to explore this challenging issue because given the scarcity of individuals with foreign experience, not all firms that would like to hire directors with this characteristic are able to do so. Furthermore, the policies to attract highly skilled returnee migrants led to an (exogenous) increase in the supply of individuals with these skills that can join the board in some provinces, but not in others.

Since the labor market for board directors is largely local (Knyazeva, Knyazeva and Masulis, 2011), as shown in Figure 1 and, more formally, in Table 4 later on, in the years following the adoption of the policies, the number of board members with foreign experience indeed increases for firms in those provinces, but not for similar firms elsewhere. The policies can therefore be used to construct instruments that are relevant.

The supply shocks determined by the policies help us identify the causal effect as long as firms are not believed to experience contextual shocks, independently affecting their performance and corporate policies. Since the policies were not targeted to listed companies and their boards, there is no reason to believe that the adoption of the policies occurs contextually to a change in the firms' demand for directors with foreign experience.

Our instruments based on the policy changes could fail to satisfy the exclusion restrictions, however, if the firms in the provinces adopting the policies earlier in our sample differed along unobserved dimensions related to firm performance from the control firms and from the firms in provinces adopting the policies later on. We address this concern by including province fixed effects (or, in alternative specifications, firm fixed effects). While this approach allows us to control for the time-invariant characteristics of the firms incorporated in a province, unobserved time-varying heterogeneity across provinces could still bias our estimates. For instance, provincial governors could issue policies in expectation of a change in the province's growth opportunities or demand for talent.

We address these concerns in several ways. First, we explore not whether firms perform better after the adoption of the policies, but whether they perform better than the median firm in their industry during the same year. Second, we design a test that exploits only *within* province (and within industry) variation as follows. After the introduction of the new policies, firms with certain *ex ante* characteristics (which we identify in Section 5.1) are more likely to attract individuals with foreign experience to their boards. We can thus test whether firms with *ex ante* characteristics that make them more likely to attract individuals with foreign experience to their boards grow more than the median listed firm within the province after the policy change (and therefore after the increase in the supply of individuals with foreign experience).

*Ex ante* firm characteristics, which we measure at the beginning of the sample period, are unlikely to predict future changes in firm performance, because we control in the second stage estimation for the contemporaneous firm characteristics. The identifying assumption is just that the effect of these firm characteristics on firm performance should not vary contextually to the policy change.

Taken jointly, these tests, together with the analysis of the changes in corporate policies that are generally set by the board, allow us to gauge the extent to which directors with foreign experience affect firm performance. We want to stress that our identification strategy allows us to identify the effect of directors with foreign experience from firms that increase the proportion of directors with foreign experience subsequently to the policy change. As Imbens and Angrist (1994) highlight in a similar context, firms in the provinces adopting the policies that are unable or unwilling to hire directors with foreign experience do not contribute to the identification (as in fixed effects estimates).

Thus, our estimates are valid as long as the control group includes predominantly firms that have similar growth opportunities and are as inclined to hire directors with foreign experience as the firms that actually hire them. For this reason, as discussed in Section 7, we explore to what extent our estimates vary if we limit the control sample by using propensity scores or, even more restrictively, by dropping firms that have no directors with foreign experience throughout the sample period.

#### 4. DATA AND SAMPLE CHARACTERISTICS

#### 4.1. Data Sources and Sample Construction

We hand-collect information on foreign education and work experience of the executive and non-executive directors of all non-financial companies in mainland China that are publicly traded on the A-share market during 1999-2009.<sup>4</sup> We manually collect the directors' bios from sina.com.cn finance and the companies' annual reports. We screen 33,707 directors' bios and cross-verify the information obtained from the bios through various news and internet searches. In this way, we obtain information on any academic degrees that the board member obtained abroad, on the academic institution granting the degree, on whether the director has worked abroad, and on the country where the director studied or worked. To ensure that foreign experience captures actual exposure to a foreign environment, we do not consider Chinese individuals who worked for a foreign branch of a Chinese company or for a Chinese branch of a foreign company or joint venture as having foreign (work) experience.

We consider only directors affiliated with companies with basic accounting and market information. We extract accounting information, stock prices, and the number of board

<sup>&</sup>lt;sup>4</sup> Chinese firms may issue three categories of shares: A shares, B shares, and H shares. A shares were originally issued for domestic investors, but since 2002 also foreign investors have been allowed to purchase them. B shares were originally issued for foreign investors; however, since March 2001, also domestic investors can hold B shares. Lastly, a limited number of firms can issue H shares on the Hong Kong Stock Exchange. In our sample, there are 52 firms with H shares. Before the 2005-2006 ownership reform, Chinese firms also issued non-tradable shares, which were held by the government and other domestic institutions.

members from the China Stock Market & Accounting Research Database (CSMAR), developed by GTA Information Technology, one of the major providers of Chinese data. After excluding firms with missing financial statements, our sample includes 1,738 firms for a total of 14,425 firm-year observations. We then apply the following filtering criteria. We first exclude 66 firm-year observations for which we have missing observations for sales, net income, number of employees, and market capitalization, and for which we are unable to compute the firm performance proxies that we describe below. We next exclude 12 firm-year observations for firms whose board has fewer than 2 directors or firms with missing information on the number of directors. We further exclude 16 firm-year observations with missing industry information. Our final sample consists of 1,733 unique firms and 14,331 firm-year observations; at the director level, we have a total of 33,707 unique directors and 138,092 individual-firm-year observations.

From the CSMAR database, we also obtain information on the tenure of board members, on the top 10 shareholders, on private placements, and on mergers and acquisitions. To define foreign ownership, we identify whether any of the top 10 largest shareholders is foreign through various news and internet searches. Our definition of foreign ownership includes foreign institutional, corporate, and individual investors, but excludes foreign branches of Chinese firms. In the same way, we establish the presence of foreign investors in private placements and foreign bidders or targets in merger and acquisitions.<sup>5</sup> From CSMAR, we also obtain information on CEO turnover from 2000 to 2009.

We gather information on firms' foreign sales from the Supplement Information on Sales in the annual reports starting from 2000.<sup>6</sup> Firms generally provide information on the regional breakdown of their sales. When a firm discloses its sales with regional breakdown

<sup>&</sup>lt;sup>5</sup> Private placements became common after 2006. Therefore, our sample period for private placements is 2006-2009.

<sup>&</sup>lt;sup>6</sup> CSMAR began reporting the Supplement Information on Sales in 2002. We manually collect data on foreign sales for 2000-2001. Most of our sample firms did not disclose their sales by region in 1999. Therefore, the sample period for the foreign sales is 2000-2009.

and does not report any sales outside mainland China, we code the firm's foreign sales as zero. If a firm does not disclose the regional breakdown of its sales, we code the firm's foreign sales as missing.

Finally, we obtain information on firms' industries, block ownership, and government ownership from the CCER China Economic and Financial Database, managed by SinoFin Information Services. Firms are classified as state-owned if their largest ultimate shareholders are either the central or a provincial government.

#### 4.2. Descriptive Statistics

Table 1 shows the number of firms and the number of firm-years affected by the policies. For instance, our sample includes 110 unique firms (831 firm-year observations) headquartered in Beijing; 161 unique firms (1,499 firm-year observations) headquartered in Shanghai; 229 unique firms (1,647 firm-year observations) headquartered in Guangdong. Importantly, the number of firms in different cities and provinces is such that each year we have a large sample of firms that are unaffected.

Panel A of Table 2 presents the main characteristics of the director level dataset. About 6.3 percent of the observations (or 8,758 director-firm-year observations) involve directors with some foreign education; of these, 2.9 percent or 4,014 director-firm-year observations refer to directors that made short-term visits, short-term training or post-docs in foreign academic institutes; 879 director-firm-year observations refer to directors with foreign bachelor degrees; 2,350 director-firm-year observations refer to directors with foreign master degrees; and 1,515 director-firm-year observations refer to directors with foreign doctoral degrees.

Most of the directors earned their highest foreign degree in the US (3,424 director-firmyear observations), followed by the UK (1,042 director-firm-year observations), Japan (924 director-firm-year observations), Hong Kong (680 director-firm-year observations), Canada (468 director-firm-year observations), and Germany (426 director-firm-year observations). A considerable number of directors obtained degrees in a variety of other countries, such as Australia, Singapore, and Sweden. Consequently, it is a director's foreign experience, rather than exposure to some specific culture or language, which is more likely to explain our findings.

Besides foreign experience, we collect information on other characteristics of board members that are generally used in the literature. Starting from 2005, CSMAR reports a flag for directors that have positions in other companies. This flag indicates that most of the directors in our sample, and especially the directors with foreign experience, have another occupation besides being board members.

We report two alternative definitions of independent director. The first one captures whether the director is also an employee of the firm (because he or she receives a salary from that firm). Slightly over 60 percent of the directors can be defined independent according to this definition. The second definition of independence is more restrictive and, as in the US post Sarbanes-Oxley, aims to capture individuals who are not affiliated with the firm, except as directors, and do not have a relationship with the company that would interfere with their "independent judgment". According to the second definition, executives of the parent company or individuals who obtain fees as consultants are not considered independent.<sup>7</sup> About 27 percent of the director-firm-year observations are independent.

Panel B of Table 2 describes the firm-level dataset, which is at the center of the empirical analysis. We start by listing our firm performance proxies: the market to book ratio (MTB); a measure of firm profitability, the ROE; and the total factor productivity (TFP).<sup>8</sup> As is common in the literature (see, for instance, Schoar, 2002), we compute a firm's total factor

<sup>&</sup>lt;sup>7</sup> China did not enforce the presence of independent directors until 2001.

<sup>&</sup>lt;sup>8</sup> We censor extreme values in the firm performance proxies as detailed in the appendix. The censoring, however, does not affect our results.

productivity as the residual,  $\hat{e}_{it}$ , of the following firm level regression  $y_{ijt} = \alpha_{jt} + \beta_{jt}l_{ijt} + \gamma_{jt}k_{ijt} + \delta_{jt}m_{ijt} + \epsilon_{ijt}$ , where  $y_{ijt}$  are the logarithm of sales of firm *i* belonging to industry *j* during year *t*,  $l_{ijt}$  is the logarithm of the number of workers of firm *i* during year *t*,  $k_{ijt}$  is the logarithm of the total assets of firm *i* during year *t*, and  $m_{ijt}$  are the logarithm of the expenses for material and other inputs of firm *i* during year *t*. We estimate all equations by industry and year. For this reason, our estimate of total factor productivity captures a firm's deviation from the factor productivity within its industry in a given year.

We then present the firm-level variables capturing board expertise. While in the empirical analysis we almost exclusively rely on the proportion of board members with foreign experience, we also report their number and a dummy variable taking a value of one if the firm has at least one director with foreign experience and taking a value of zero otherwise. Approximately 46 percent of the observations in our sample refer to firm-years in which firms have at least one director with foreign experience.<sup>9</sup>

Our dataset contains information on board structure and ownership structure. On average, Chinese firms have slightly less than 10 board members. Thus, board size is slightly smaller than in the U.S. where on average, listed companies have about 12 directors (Yermack, 1996).

We control for the percentage ownership of the largest shareholder. Existing literature (e.g., McConnell and Servaes, 1990; Morck, Shleifer and Vishny, 1988) highlights that ownership concentration may improve performance, because it strengthens shareholders' incentives to monitor. Government ownership is another important control as central and local governments own shares in nearly 70 percent of the firm-year observations in our sample.

<sup>&</sup>lt;sup>9</sup> In unreported specifications, we also distinguish between directors with foreign experience that are directly involved in management (such as CEOs and Chairmen of the Board) and other directors with foreign experience. While our findings are qualitatively invariant, we do not find any differences between the two, highlighting the importance of both the advising and the monitoring role of the board.

Last, but not least, we control for foreign ownership. Foreign blockholders are believed to bring superior technology, organizational capital, and access to international capital markets (see for instance, Chari, Chen and Dominguez 2009; Desai, Foley and Forbes, 2007; Haskel, Pereira and Slaughter, 2007). Regulatory restrictions on foreign ownership of listed companies and on the activities of foreign financial institutions, however, significantly limit the role of foreign owners in Chinese listed companies. This is also apparent from the fact that foreigners own about 2 percent of the stocks of our sample firms. Nevertheless, to the extent that firms with a foreign blockholder are more likely to engage directors with foreign experience, without including this control, we could erroneously ascribe changes in performance due to changes in foreign ownership to changes in board attributes.

Panel C of Table 2 reports the industry distribution of firms with directors with and without foreign experience. While all industries have firms with directors with foreign experience, the industries in which more firms do so are machinery, gas and chemistry, metal, information technology and pharmaceutical products. Unsurprisingly, these are industries in which scientific knowledge acquired abroad may play a particularly important role.

Table 3 compares the characteristics of firms that eventually hire directors with foreign experience in the years before they have done so with those of firms that do not hire directors with foreign experience during the sample period, our (main) control group. In comparison to the control group, firms that end up hiring directors with foreign experience appear to have similar growth opportunities (market to book ratio), profitability (ROE) and conduct a similar number of foreign M&A transactions and private placements, indicating that there are no pre-existing differences. However, firms that end up hiring directors with foreign sales. These firms are also smaller,

have a slightly larger board size, a slightly lower leverage, are more likely to have the government or a block holder as shareholders.

To take into account these *ex ante* differences, we control for these firms characteristics as well as for industry or firm fixed effects throughout the analysis. In some robustness tests, we also restrict the control sample to firms that are *ex ante* more similar to the ones with foreign directors either using propensity scores or focusing only on firms that end up hiring directors with foreign experience during the sample period.

#### **5. RESULTS**

#### 5.1. Which Firms Have Directors with Foreign Experience?

Table 4 relates the proportion of directors with foreign experience to firm characteristics and alternative variables capturing the provincial policies to attract highly skilled emigrants.<sup>10</sup> It shows that firms with a greater proportion of directors with foreign experience have a higher level of foreign ownership and are less likely to have the government as shareholder. It also appears that these firms are larger and more profitable. However, once we control for firm fixed effects, in column 2, board structure seems to be affected only by ownership characteristics and firm size.<sup>11</sup> The effect of foreign ownership is not only statistically significant but also economically significant: A one-standard-deviation increase in foreign ownership increases the proportion of directors with foreign experience by nearly 4 percentage points.<sup>12</sup>

<sup>&</sup>lt;sup>10</sup> Even though our dependent variable is truncated at zero and one, here, we estimate parameters by ordinary least squares instead of using a Tobit model. The Tobit estimator relies on the distributional assumptions and is inconsistent when disturbances are non-normal (Arabmazar and Schmidt, 1982). In contrast, in a standard linear regression model, the ordinary least square estimator is unbiased and consistent even when the assumption of normality of the disturbances is violated.

<sup>&</sup>lt;sup>11</sup> Importantly, in unreported results, we do not find that the percentage of directors with foreign experience is followed by an increase in foreign ownership.

<sup>&</sup>lt;sup>12</sup> Results are qualitatively similar if instead of the proportion of directors with foreign experience, we use the number of directors with foreign experience (and control for board size) or, alternatively, a dummy variable that takes a value of one if the firm has at least one director with foreign experience. For brevity, we do not tabulate these results.

Starting from column 3, we explore the role of the provincial policies. We construct a dummy variable that takes a value of one for firms incorporated in the province in the years following the adoption of the policy. As expected, the dummy capturing the timing of the policy changes has a positive and statistically significant effect on the proportion of board members with foreign experience.

In column 4, we consider how the effect of the policy differs across firms with different ownership structures. It appears that firms that have a foreign blockholder or the government as shareholders are more likely to be able to attract directors with foreign experience after the implementation of the policy (although the effect of the interaction of the policy dummy with the dummy capturing government ownership is not statistically significant at conventional levels).

The effect of the policies we highlight is in excess of a province-specific (linear) trend, indicating that after the policy adoption the proportion of directors with foreign experience increases faster than during previous years in that province, and it is highly significant even if we cluster errors at the year level to account for eventual common shocks leading to a higher proportion of directors with foreign experience. This indicates that the timing of the policies adoption can be used to construct instruments for the proportion of directors with foreign experience. The Staiger-Stock (1997) F-statistics, which we present in columns 3 and 4, reveals that the instruments based on the policies (that is, the policy dummy in column 3 and the policy dummy and the interaction of the policy dummy with firm ownership characteristics in column 4) can be strong instruments for the proportion of directors with foreign experience.

Below, we construct instrumental variables for the proportion of directors with foreign experience using the policy dummy. As explained in Section 3, to be able to design the within province tests, we further instrument the proportion of directors with foreign

18

experience with interactions of the policy dummy and the *ex ante* firm ownership characteristics. We use the percentage of foreign ownership, state ownership and the fraction of shares held by the largest shareholder at the beginning of the sample period. In the second stage estimation, we control for the contemporaneous effect of these firm characteristics.

#### 5.2. Directors with Foreign Experience and Firm Performance

We study whether a larger proportion of directors with foreign experience is associated with better firm performance. Table 5 focuses on corporate valuations, as measured by the market-to-book ratio (MTB). In all columns but 6, we define the MTB in deviation from the industry-year median by subtracting from the firm's MTB the industry median MTB in year *t*. In this way, we capture the ranking of firms within an industry and are able to abstract from industry shocks.<sup>13</sup>

In columns 1 through 3, we present the ordinary least square estimates, first, using a limited set of controls, then, controlling for ownership structure, and finally, including also firm fixed effects. Although the parameter estimate of our variable of interest becomes smaller especially when we add firm fixed effects, we always find a positive effect of a higher proportion of directors with foreign experience on performance.

In columns 4 to 6, we present the instrumental variable estimates. In all cases, the estimates indicate a positive and significant effect of a higher proportion of directors with foreign experience on firm valuations. In particular, the results cannot be interpreted to depend on differences in growth opportunities across industries as the dependent variable is defined in deviation from the MTB of the median firm in the industry in year *t*. In column 5, we include province fixed effects, which account for time-invariant firm heterogeneity across

<sup>&</sup>lt;sup>13</sup> Our results do not depend on the fact that we use industry-adjusted variables. In unreported specifications, we estimate the ordinary least squares specifications by including industry times year effects and/or province times year effects. The results are similar to the ones we report.

provinces. Results are similar if, instead of the province fixed effects, we include firm fixed effects.

In other (unreported) specifications, we control for the aggregate GDP growth in the province (instead of the province fixed effects), to take into account time-varying growth opportunities across provinces. Both our ordinary least squares and instrumental variable estimates are similar to the ones we report.

In columns 6, we account for differences in time-varying growth opportunities across provinces by subtracting from the firm's MTB the median MTB of the firms in the same province in year *t* as well as the median MTB of the firms in the same industry in year *t*. The estimates continue to imply a positive effect of the directors with foreign experience on firm valuation and indicate that precisely the firms that are more likely to attract directors with foreign experience perform better after the introduction of the policy. If anything, fully accounting for time-invariant heterogeneity increases the magnitude of the coefficient of our variable of interest.

Importantly, the effects we highlight are not only statistically significant, but also economically large, especially when we exploit the exogenous variation in the proportion of directors with foreign experience. Using the ordinary least square estimates in column 3, a one-standard deviation change in the proportion of directors with foreign experience leads to a 0.04-standard deviation change in the dependent variable, an overall tiny number.<sup>14</sup> By contrast, in column 5, when we consider only the variation in the proportion of board members with foreign experience due to the provincial policies, a one-standard deviation change in the proportion of directors with foreign experience sufficient to bring a firm with median valuation to nearly the 80<sup>th</sup> percentile.

<sup>&</sup>lt;sup>14</sup> We obtain the economic magnitude of the coefficients using the standard deviation of the industry-yearadjusted market to book ratio, which is 1.195.

This large increase in the parameter estimates has an intuitive economic interpretation. The proportion of directors with foreign experience captures *all* directors with foreign experience. However, directors have different foreign experiences and some of these experiences can certainly be more valuable for firms than others. Since the policies were directed only to individuals with the most distinguished backgrounds, when we exploit the exogenous variation, we concentrate on the most skilled individuals, who can make the most valuable contributions to firm performance. It is therefore natural that we obtain stronger results.

Table 6 and Table 7 repeat the same set of exercises for two other measures of firm performance, total factor productivity and profitability, respectively. Since we expect any effects on accounting profits to be delayed, we consider profitability not during the year in which the policy is implemented, like for the other performance measures, but one year afterwards. The estimates are still strongly supportive of a positive effect of directors with foreign experience on performance. For instance, based on column 5 in each table, a one-standard-deviation increase in the fraction of directors with foreign experience can bring a firm with median productivity to the 65<sup>th</sup> percentile and a firm with median ROE to have an ROE slightly above the 75<sup>th</sup> percentile.

#### 6. WHAT DO DIRECTORS WITH FOREIGN EXPERIENCE DO?

The causal mechanism behind our maintained hypothesis that directors with foreign experience positively affect firm performance implies that the way in which firms are run changes when directors with foreign experience join the board. We thus explore whether directors with foreign experience affect policies that are a prerogative of the board, such as firm internationalization and corporate governance. First, we ask whether the probability that the firm does an international merger or acquisition, as opposed to a domestic deal, is larger when a higher proportion of the firm's board members have foreign experience. As before, we present ordinary least squares and instrumental variable estimates. The estimates in columns 1 and 2 in Panel A of Table 8 indicate that this is the case. In column 3, we interact the proportion of directors with foreign experience and a dummy for whether the target firm is from the same country in which any of the directors obtained the foreign experience. The coefficient of the interaction term is positive and highly significant, fully supporting the causal interpretation of the empirical evidence.

Second, we consider capital raising activities and in particular whether firms do private placements with international or domestic investors.<sup>15</sup> By considering only firms that do a private placement, we keep the demand for equity constant. We also control for whether a firm has issued B shares and/or H shares in addition to A shares, as a firm's ability to engage foreign investors depends on whether foreign investors can trade its shares. Our data on private placements starts in 2006, which significantly reduces the sample size and explanatory power of the instruments. Nevertheless, columns 4 and 5 in Panel A of Table 8 indicate that a higher proportion of directors with foreign experience increases the probability that the firm makes a private placement with a foreign investor.<sup>16</sup>

Third, we examine the impact of a larger portion of directors with foreign experience on firms' foreign sales. Columns 6 and 7 in Panel A of Table 8 provide the OLS and IV estimates. A higher proportion of directors with foreign experience is associated with a larger fraction of sales outside mainland China in comparison to other firms in the same industry during that year. Also in this case, the coefficient of the fraction of directors with foreign

<sup>&</sup>lt;sup>15</sup> Our sample includes few cross-listed firms, because our sample firms are traded in the Chinese A-share markets, which imply that their first listing was in mainland China. Chinese cross-listed firms generally list first in the foreign market, and only relatively few of them afterwards obtain a listing in China.

<sup>&</sup>lt;sup>16</sup> We present the OLS and IV results so that the coefficients are comparable. In unreported tests, we estimate logit models for foreign mergers and acquisitions and for foreign private placement. We find similar results.

experience is larger when we use instrumental variables. Overall, these findings suggest that board members with foreign experience help to shape the international activity of the firm.

Lastly, in Panel B of Table 8, we consider two aspects of corporate governance, earning management and CEO turnover. Better governed firms should be more transparent and manage their earnings to a lower extent. Following Stubben (2010), we construct a proxy for earnings management from the residuals of a regression for discretionary revenues that we run for each industry and year.<sup>17</sup> In columns 1 and 2 of Panel B, we find that a higher proportion of directors with foreign experience is associated with a lower degree of earnings management.

Next, in column 3 of Panel B, we estimate how the presence of directors with foreign experience affects the probability of CEO turnover. A higher proportion of directors with foreign experience increases the probability of CEO turnover in firms with relatively poor performance, as the negative and significant coefficient of the interaction term between the proportion of directors with foreign experience and the firm's ROA indicates. Overall, the estimates suggest that directors with foreign experience facilitate firm internationalization and improve corporate governance.

#### 7. ROBUSTNESS

#### 7.1. Directors or Other Employees?

A potential alternative explanation of our results is that, while the policies to attract talented emigrants with foreign experience indeed had a positive impact on firm performance, this may have affected firms also through other channels, such as the hiring of returnee emigrants in positions other than the board of directors. Such an explanation would be broadly consistent with the positive effect of human capital we highlight, especially if

<sup>&</sup>lt;sup>17</sup> For this reason, in these specifications, we do not include industry fixed effects and do not subtract from the dependent variable the industry-year median.

directors with foreign experience facilitate the recruitment of a few other employees with foreign experience.

A generalized change in the labor force of all firms affected by the provincial policies is instead unlikely to explain our findings. First, as we show in Section 6, corporate policies that are typically set by the board, and not by lower level employees, change after the arrival of directors with foreign experience.

Second, the highly talented returnee emigrants benefiting from the policies are a tiny proportion of the Chinese population and the workforce of each province (during the sample period, the number of returnee emigrants goes from less than 5,000 to around 30,000 in a population of over 1.3 billion and only a small fraction of these were sufficiently highly skilled to benefit from the policies). Thus, the policies we consider cannot have changed the composition of the labor force of the large listed companies in our sample. Even if firms hired a few individuals with foreign experience in non-director positions, these could plausibly have a tiny effect on the firm's overall performance, which we would be unlikely to be able to uncover. Directors affecting the firm's strategy and corporate governance can more plausibly generate the effects that we highlight. Consistently, with this explanation, in unreported specifications, we do not find any changes in the total wages paid by the sample firms to lower level employees, which should presumably reflect (large) changes in the labor force composition of the firm.<sup>18</sup>

Nevertheless, we test whether the policies matter predominantly through the composition of the board of directors as follows. We ask whether the adoption of a policy for attracting highly talented returnee emigrants affects all firms in the province or rather only the firms with a non-zero proportion of directors with foreign experience, as we would expect if the impact of the policies were exclusively through the composition of the board of

<sup>&</sup>lt;sup>18</sup> This is not inconsistent with the fact that, as we discuss in Subsection 7.4, we do not find an effect on director compensation, as being a board member of a listed company is a prestigious position *per se* and does not involve a full time commitment.

directors. Our estimates in Table 9 clearly indicate that the effect of the policies is through the composition of the board. While the policies appear to have a positive effect on all three measures of performance for firms that end up having at least one director with foreign experience, the effect for the remaining firms in the province is negative and is not statistically different from zero, indicating that the directors' foreign experience is crucial.

#### 7.2. Foreign Experience vs. Other Board Characteristics

So far we have shown that directors with foreign experience affect firm performance and corporate policies. However, it is possible that these highly skilled returnee emigrants change the structure of the board along other dimensions that are known to affect performance. For this reason, we explore whether the proportion of directors with foreign experience is still positively related to performance if we control for other characteristics of the board. We include controls for the following board characteristics: the average tenure of the directors, the proportion of board members who are also employed in the firm,<sup>19</sup> board size, the average age of board members, the proportion of female directors, the proportion of directors that attended and/or worked for either Tsinghua University or Peking University, which are considered the Chinese most selective universities, the proportion of foreign board members, the proportion of board members with political connections, and the proportion of busy directors.

Tenure and board independence are sometimes believed to capture board entrenchment and may thus characterize boards that are less effective in performing their monitoring function. Others, however, argue that directors with longer tenures or who hold posts in the firm may have better information and are more effective advisors. Since, given the nature of

<sup>&</sup>lt;sup>19</sup> Since China did not enforce the broader definition of independent directors until 2001, for the whole sample, we can only control for board independence using the fraction of directors employed by the firm, an inverse measure of board independence.

our experiment, directors with foreign experience have shorter tenure, it is important to control for the average tenure of the board members.

Similarly, board size is often considered to influence the effectiveness of the board. Some argue that large boards are less effective because of coordination problems. Others object that firms with large boards are able to draw from a broader range of expertise. The arrival of directors with foreign experience may increase board size. To the extent that boards become too large, the negative effect of board size may attenuate the positive effects of their expertise on firm performance.

By the same token, some associate busy directors (that is, directors who belong to the corporate board of more than one firm) with ineffective monitoring (Core, Holthausen and Larcker, 1999; Fich and Shivadasani, 2006), while others argue that busy directors have more experience and better connections, and thus can help improve firm value (Field, Lowry and Mkrtchyan, 2011). To the extent that directors with foreign experience are busier, this could also affect our findings.

Demographic characteristics are often included as controls in studies exploring the effect of board structure on firm performance. In particular, the proportion of directors that attended the Chinese most selective universities allows us to control for the extent to which the proportion of directors with foreign experience may be correlated with the directors' innate abilities and quality of education, rather than with their foreign experience.

Masulis, Wang and Xie (2012) show for US listed companies that foreign individuals may bring expertise. However, because of physical distance and cultural differences, they appear not to be effective monitors, even in a high transparency environment like the US, and are in fact extremely rare (especially, in our sample). Consistently with this evidence, our estimates indicate that foreign directors are not as beneficial for firm performance as the Chinese directors with foreign experience.

26

Furthermore, board members' political connections are particularly important in the Chinese context. Fan, Wong and Zhang (2007) show that Chinese companies with politically connected CEOs are more likely to appoint bureaucrats, rather than professionals, to the board and underperform those without politically connected CEOs. The concern may thus arise that board members with foreign experience are hired by the most efficient companies without political connections and that this may be driving our results. This is unlikely because our results are invariant when we include firm fixed effects and when we control for state ownership, a proxy for formal political connections. Furthermore, the empirical evidence on the effect of political connections on firm performance is mixed. Calomiris, Fisman and Wang (2010) find a positive effect of political connections on firms' valuations. Similarly, Cao et al. (2011) find that CEOs with political career concerns improve firm performance. Nevertheless, we further address this criticism and control for the political ties of the directors by including the proportion of politically connected board members, which, following Fan, Wong and Zhang (2007) and Calomiris, Fisman and Wang (2010), we define as directors that are currently or were previously employed as bureaucrats by the central or by a local government.

Table 10 shows that the effect of board members with foreign experience is unchanged when we control for other characteristics of the board. For brevity, we report only the instrumental variable estimates.

#### 7.3. Propensity Scores

Our main analysis is based on the whole sample of publicly traded companies. Since individuals with foreign experience are a relatively scarce resource in China, our control sample largely consists of firms that would like to employ directors with foreign experience, but are unable to do so for idiosyncratic reasons, even though they are similar to firms that have directors with foreign experience. To further refine our control sample and check the robustness of our findings, we use a propensity score framework.

For firms employing directors with foreign experience, we use a nearest neighbor matching estimator based on year, industry, size and leverage to create a matching sample of firms without directors with foreign experience. We then compare the performance of matched firms with and without directors with foreign experience by re-estimating Tables 5 to 7. The results in columns 1 to 3 of Table 11 show that our estimates are unaffected.

In other unreported tests, we re-estimate Tables 5 to 7 excluding any firms that never have directors with foreign experience during the sample period. Also in this case the estimates remain qualitatively and quantitatively similar to the ones we present.

#### 7.4. More Evidence on (the Absence of) Selection Effects

Could firms that attract directors with foreign experience differ along dimensions that independently lead to the changes in performance and corporate policies we have highlighted? The tests we have presented so far indicate that this is unlikely and, in particular, that differences across provinces do not drive our findings. In this section, we provide further evidence that other types of firm heterogeneity are also unlikely to drive our findings.

In columns 4 to 6 of Table 11, we re-estimate the main equations in Tables 5, 6 and 7 including firm fixed effects and the lagged dependent variable as a control. For brevity, we report only the instrumental variable estimates. It is apparent that directors with foreign experience, if anything, have a larger effect in firm performance when we fully absorb firm time-invariant heterogeneity and control for any time-varying heterogeneity captured by the firm's previous performance.

Further, independent evidence supporting our identification strategy comes from the analysis of the compensation of board members. We surmise, as general equilibrium models of the allocation of talent imply (e.g., Murphy and Zabojnik, 2004), that if directors with foreign experience sort in firms with higher demand for their skills, then these firms should pay more for their services.

In unreported tests, we find unambiguous evidence that this is not the case. Firms with directors with foreign experience do not pay (all) their directors more. Moreover, within these firms, directors with foreign experience are not paid more than other directors. Thus, this evidence indicates, consistently with our previous tests, that no bidding process occurs and that directors with foreign experience do not sort across firms on the basis of firms' unobservable characteristics. In particular, directors with foreign experience are unlikely to have joined companies in which their skills are needed more than in others. It rather appears that the widespread scarcity of talent in China makes the control sample of firms that do not employ directors with foreign experience highly comparable.

#### 7.5. The Quality of Foreign Experience

In other tests that we do not report for brevity, we distinguish between foreign work experience and education. If we focus on foreign work experience, we find that it increases firm performance. However, if we run a horse race between foreign education and foreign work experience, the effect of foreign education appears to be more important than the effects of foreign work experience. This may depend on the fact that individuals with foreign work experience also have foreign education and that a foreign education allows to access better foreign work experience.

In Table 12, we compare the effect of (any) foreign education and of foreign education that led to a foreign academic degree. In all cases, we find that the coefficient of foreign education that led to an academic degree is larger and more precisely estimated (although the difference between foreign experience and foreign experience that led to an academic degree is statistically significant only for TFP), further supporting the causal interpretation of our findings.

#### 8. CONCLUSIONS

The brain drain from emerging markets may not only have costs, but also positive, indirect, benefits. Talented individuals migrating to foreign countries accumulate knowledge and skills. If these highly skilled emigrants ever decide to return, the experience they gained abroad can benefit their home country and the brain drain becomes a brain gain.

This paper documents a specific channel leading to brain gain. We show that when individuals with foreign experience join corporate boards, firm performance improves and firms are run differently. The positive causal effects on firm valuations are large relative to the compensation of board members, which is moderate in China, and suggest that the externalities created by highly skilled returnee emigrants in emerging economies are potentially large.

We argue that the positive effects on firm performance we document depend on the adoption of superior management practices, improvements in corporate governance and easier access to foreign investors and technologies through M&As. The externalities created by highly skilled returnee emigrants are unlikely to be confined to the universe of listed companies and are likely to be found, and potentially be even larger, also in entrepreneurial firms or universities and education, the main target of the policies. We consider this an exciting area for future research.

#### REFERENCES

Adams, R. B., B. E. Hermalin and M. S. Weisbach, 2010, The Role of Boards of Directors in Corporate Governance: A Conceptual Framework and Survey, *Journal of Economic Literature* 48, 58-107.

Agrawal, A., and S. Chadha, 2005, Corporate Governance and Accounting Scandals, *Journal of Law and Economics* 48, 371-406.

Ahearn, K., and A. Dittmar, 2011, The Changing of the Boards: The Impact on Firm Valuation of Mandated Female Board Representation, *Quarterly Journal of Economics* forthcoming.

Arabmazar, A., and P. Schmidt, 1982, An investigation of the robustness of the Tobit estimators to non-normality, *Econometrica* 50, 1055-1069.

Bhagwati, J., 1974, The Brain Drain, International Integration of Markets for Professionals and Unemployment: A Theoretical Analysis, *Journal of Development Economics* 1, 19-42.

Bloom, N., and J. V. Reenen, 2007, Measuring and Explaining Management Practices across Firms and Countries, *Quarterly Journal of Economics* 122, 1351-1408.

Bloom, N., B. Eifert, A. Mahajan, D. McKenzie, and J. Roberts, 2012, Does management matter? Evidence from India, *Quarterly Journal of Economics* forthcoming.

Calomiris, C. W., R. Fisman, and Y. Wang, 2010, Profiting from government stakes in a command economy Evidence from Chinese asset sales, *Journal of Financial Economics* 96, 399-412.

Cao, J., M. L. Lemmon, X. Pan, M. Qian, and G. G. Tian, 2011, Political Promotion, CEO Incentives, and the Relationship between Pay and Performance, Working Paper, University of Utah.

Chhaochharia, V., and Y. Grinstein, 2009, CEO compensation and board structure, *Journal of Finance* 64, 231–261.

Chari, A., W. Chen, and K. M. E. Dominguez, 2011, Foreign Ownership and Corporate Restructuring: Direct Investment by Emerging-Market Firms in the United States, Working Paper, University of North Carolina at Chapel Hill.

Coles, J. L., N. D. Daniel, and L. Naveen, 2008, Board: Does one size fit all? *Journal of Financial Economics* 87, 329-356.

Core, J., R. Holthausen, and D. Larcker, 1999, Corporate governance, chief executive officer compensation, and firm performance, *Journal of Financial Economics* 51, 371-406.

Desai, M. A, C. F. Foley, and K. J. Forbes, 2007, Financial Constraints and Growth: Multinational and Local Firm Responses to Currency Depreciations, *Review of Financial Studies* 19, 1433-1464.

Docquier, F. and H. Rapoport, 2012, Globalization, Brain Drain, and Development, *Journal of Economic Literature* 50, 681–730.

Fama, E. F., and M. C. Jensen, 1983, Separation of ownership and control, *Journal of Law and Economics* 26, 301–325.

Fan, J., T. Wong, and T. Zhang, 2007, Politically connected CEOs, corporate governance, and Post-IPO performance of China's newly partially privatized firms, *Journal of Financial Economics* 84, 330–357.

Farrell, D., and A. J. Grant, 2007, China Looming Talent Shortage, McKinsey, New York.

Fich, E, and A. Shivasani, 2006, Are busy boards effective monitors? *Journal of Finance* 61, 689-724.

Field, L., M. Lowry, and A. Mkrtchyan, 2011, Are Busy Boards Detrimental? Working Paper, Penn State University.

Green, S., 2003, China's Stockmarket: A Guide to its Progress, Players and Prospects, The Economist, London.

Guadalupe, M., O. Kuzmina, and C. Thomas, 2012, Innovation and foreign ownership, *American Economic Review* forthcoming.

Gul, F. A., J. Kim, and A. A. Qiu, 2010, Ownership concentration, foreign shareholding, audit quality, and stock price synchronicity: Evidence from China, *Journal of Financial Economics* 95, 425-442.

Guthrie, K., J. Sokolowsky, and K. M. Wan, 2012, CEO compensation and board structure revisited, *Journal of Finance* forthcoming.

Güner, A. B., U. Malmendier, and G. Tate, 2008, Financial Expertise of Directors, *Journal of Financial Economics* 88, 323–354.

Hall, R., and C. Jones, 1999, Why Do Some Countries Produce So Much More Output per Worker Than Others? *Quarterly Journal of Economics* 114, 83–116.

Haskel, J., S. Pereira, and M. Slaughter, 2007, Does Inward Foreign Direct Investment Boost the Productivity of Domestic Firms?, *Review of Economics and Statistics* 89, 482-496.

Imbens, G. W., and J. D. Angrist, 1994, Identification and Estimation of Local Average Treatment Effects, *Econometrica* 62, 467-475.

Jiang, W., H. Wan, and S. Zhao, 2012, The Two-Sided Career Concerns of Independent Directors: Evidence from Director Voting, Working Paper, Columbia University.

Jones, C., and P. Romer, 2009, The New Kaldor Facts: Ideas, Institutions, Population, and Human Capital, NBER Working Paper 15094.

Khanna, T., 2008, Billions Entrepreneurs. How China and India Are Reshaping Their Futures and Yours, Harvard Business School Press, Cambridge, MA.

Klein, A., 1998, Firm performance and board committee structure, *Journal of Law and Economics* 41, 275-303.

Knyazeva, A., D. Knyazeva, and R. Masulis, 2011, Effects of Local Director Markets on Corporate Boards, Working Paper, University of Rochester.

Masulis, R., C. Wang, and F. Xie, 2012, Globalizing the boardroom-the effects of foreign directors on corporate governance and firm performance, *Journal of Accounting and Economics* forthcoming.

McConnell, J. J., and H. Servaes, 1990, Additional Evidence on Equity Ownership and Corporate Value, *Journal of Financial Economics* 27, 595-612.

Morck, R., A. Shleifer, and R. W. Vishny, 1988, Management Ownership and Market Valuation: An Empirical Analysis, *Journal of Financial Economics* 20, 293-315.

Murphy, K. J., and J. Zabojnik, 2004, CEO pay and turnover: a market based explanation for recent trends, *American Economic Review* 94, 192–196.

Nanda, R. and T. Khanna, 2010, Diasporas and Domestic Entrepreneurs: Evidence from the Indian Software Industry, *Journal of Economics and Management Strategy* 19, 991–1012.

Schoar, A., 2002, Effects of corporate diversification on productivity, *Journal of Finance* 57, 2379-2403.

Staiger, D., and J. H. Stock, 1997, Instrumental variables regression with weak instruments, *Econometrica* 65, 557-586.

Stubben, S. R., 2010, Discretionary Revenues as a Measure of Earnings Management, *Accounting Review* 85, 695-717.

Stulz, R. M., 1999, Globalization, Corporate Finance and the Cost of Capital, *Journal of Applied Corporate Finance*, 12, 8-25.

The Economist, 2011, Drain or Gain?, May 26.

Wang, W., X. Zeng, and W. Pu, 2011, Guidelines for Overseas Returnees to Set Up Ventures in China (in Chinese), China Machine Press, Beijing, China.

Yermack, D., 1996, Higher market valuation of companies with a small board of directors, *Journal of Financial Economics* 40, 185-211.

Zweig, D., 2006, Competing for talent: China's strategies to reverse the brain drain, *International Labor Review* 145, 65-89.

### Appendix: Variable Definition and Construction

Variable	Definition and Data Source
Age	The difference between the current year and the director year
	of birth. Source: Manual collection and CSMAR database.
Average Age	Average age of a firm's directors. Winsorized at 1% and 99% levels.
Average Tenure	Average tenure of a firm's directors. Winsorized at 1% and 99% levels.
Assets	Total assets of the firm (in RMB 100 millions). Winsorized at 1% and 99% levels. Source: CSMAR database.
Block	Fraction of shares held by the largest shareholder. Winsorized at 1% and 99% levels. Source: CCER database.
Board Political Connection	Fraction of politically connected directors in the board. A director is defined as politically connected if he or she is a current or former government bureaucrat following Fan et al. (2007) and Calomiris et al. (2010). Winsorized at 1% and 99% levels. Source: Manual collection.
Board Size	The number of directors. Winsorized at 1% and 99% levels. Source: CSMAR database.
Busy Directors (Fraction)	Fraction of directors who sit on the boards of two or more publicly traded firms. Winsorized at 1% and 99% levels. Source: Manual collection and CSMAR database.
CEO Turnover	A dummy equal to one if there is a CEO turnover in a given year. Source: Manual collection and CSMAR database.
Directors with Foreign Experience Dummy	A dummy variable equal to one if at least one director has either foreign working experience, or foreign education, or both, and zero otherwise. Source: Manual collection.
Dummy for B/H-share	A dummy variable equal to one if a firm has B and/or H shares in addition to A shares. Source: CSMAR database.
Earnings Management	Following Stubben (2010), for each year-industry group, we regress the change in gross accounts receivable scaled by total assets at the beginning of the year on $\Delta$ sales (change in sales scaled by total assets at the beginning of the year), the interactions between $\Delta$ sales and log of total assets, between $\Delta$ sales and log of firm's age, between $\Delta$ sales and sales growth adjusted by industry-year median if the latter is greater than 0, between $\Delta$ sales and sales growth adjusted by industry-year median if the latter is smaller than 0, between $\Delta$ sales and gross profit margin adjusted by industry-year median, as well as the interaction between $\Delta$ sales and the square of the log of firm's age, and the interaction between $\Delta$ sales and the square of gross profit margin adjusted by industry-year median. We then take the absolute value of the residual. Winsorized at 1% and 99% levels. Source: CSMAR database.
Employed Directors	Proportion of directors who receive a salary from the firm.
(Fraction)	Winsorized at 1% and 99% levels. Source: Manual collection.
Female Directors (Fraction)	Proportion of female directors. Winsorized at 1% and 99% levels. Source: Manual collection.

Foreign Directors (Fraction)	Proportion of directors that are foreign nationals. Winsorized at 1% and 99% levels. Source: Manual collection.
Foreign Education	Fraction of directors with foreign education. Source: Manual collection.
Foreign Experience	Fraction of directors with foreign experience. Calculated as the number of directors that have either foreign working experience, or foreign education, or both, scaled by the total number of board directors. Winsorized at 1% and 99% levels. Source: Manual collection and CSMAR database.
Foreign M&A	A dummy equal to one if at least one of the merger & acquisition transactions announced by a sample firm in a given year involves a foreign target firm, and zero if the M&A transactions announced by a sample firm in a given year involve no foreign targets. Source: Manual collection and CSMAR database.
Foreign Ownership	Fraction of shares held by foreign investors. Winsorized at 1% and 99% levels. Source: Manual collection.
Foreign Private Placement	A dummy equal to one if at least one of the private placements filed by the firm in a given year is targeted to foreign investors, and zero if none of these private placements is targeted to foreign investors. Source: Manual collection and CSMAR database.
Foreign Sales	Foreign sales as a fraction of total sales. Winsorized at 1% and 99% levels. Source: Manual collection and CSMAR database.
Foreign Work Experience	Fraction of directors with foreign work experience. Source: Manual collection.
Leverage	Total liabilities divided by total assets. Winsorized at 1% and 99% levels. Source: CSMAR database.
MTB	Market-to-book ratio. Constructed as the sum of the market value of equity and book value of total liabilities, scaled by the book value of total assets. We censor this variable if it is above 10 or below 0. Source: CCER and CSMAR databases.
Number of Directors with Foreign Experience	The total number of directors that have either overseas working experience, or foreign education, or both. Winsorized at 1% and 99% levels. Source: Manual collection.
ROA	Operating income divided by total assets. Winsorized at 1% and 99% levels. Source: CSMAR database.
ROE	Net income divided by total equity. The sample period for this variable is 1999-2010. We censor this variable if it is above 2 or below -2. Source: CSMAR database.
Size	Natural log of total assets. Winsorized at 1% and 99% levels.
State	A dummy variable equal to one if a firm is government controlled or owned, and zero otherwise. State ownership includes central and provincial government ownership. Source: CCER database.
Tenure	One plus the difference between the current year and the year when an individual joined the firm's board of directors. Source: Manual collection and CSMAR database.

	Total factor productivity. Defined as in Schoar (2002). For all			
TFP	firms in an industry and a given year, we regress the natural			
	logarithm of sales on the natural logarithm of its total assets,			
	the natural logarithm of its total number of employees, and			
	the natural logarithm of cash payments for raw materials and			
	service. The firm's TFP is computed as the residual of this			
	regression. Winsorized at 1% and 99% levels.			
	Fraction of directors that graduated from and/or worked for			
Chinese University	Peking University and/or Tsinghua University, the two top			
(Fraction)	universities in China. Winsorized at 1% and 99% levels.			
	Source: Manual collection.			

#### Figure 1 Fraction of Directors with Foreign Experience

This figure presents the fraction of directors with foreign experience before and after the implementation of the provincial policies to encourage the return of highly skilled emigrants in the event time. The first two columns compare the fraction of directors with foreign experience before and after the implementation of the policies for firms in each of the provinces. We also compare the fraction of directors with foreign experience for the firms located in Beijing, Shanghai, and Guangdong before and after the implementation of the policies in each of these provinces.



### Table 1Policies to Attract Highly Skilled Emigrants

This table reports the year of the policy adoption, the number of unique sample firms, and the number of firm-year observations for each province that implements a policy to attract highly talented emigrants. The sample period is 1999-2009. "Issuing year" is the year when the policy was adopted. "After" refers to firm-year observations after the issuing year. "Before" refers to firm-year observations before and during the issuing year.

Drovinco	Isouin a voon	# of unique	# of firm	# of firm-year observations		
Province	issuing year	firms	Total	Before	After	
Anhui	1994	59	436	0	436	
Beijing	2000	110	831	92	739	
Chongqing	2005	30	293	180	113	
Fujian	2000	62	488	71	417	
Gansu	2003	23	202	82	120	
Guangdong	1999	229	1,647	114	1,533	
Guangxi	2005	27	232	133	99	
Guizhou	2003	18	158	56	102	
Hainan	2001	23	219	58	161	
Hebei	2001	39	347	74	273	
Heilongjiang	2002	33	330	120	210	
Henan	1992	44	346	0	346	
Hubei	2002	66	622	207	415	
Hunan	2001	53	439	90	349	
Inner Mongolia	2001	23	215	50	165	
Jiangsu	2004	127	930	391	539	
Jiangxi	2003	27	242	88	154	
Jilin	2001	39	356	90	266	
Liaoning	1999	61	554	43	511	
Ningxia	2003	11	114	49	65	
Qinghai	1999	10	99	7	92	
Shaanxi	1995	30	260	0	260	
Shandong	2005	105	813	445	368	
Shanghai	2005	161	1,499	924	575	
Shanxi	2007	26	232	183	49	
Sichuan	2005	80	685	415	270	
Tianjin	2001	28	245	49	196	
Tibet	N/A	9	80	N/A	N/A	
Xinjiang	2003	34	278	103	175	
Yunnan	2001	27	234	51	183	
Zhejiang	2001	135	905	143	762	

# Table 2Summary StatisticsPanel A: Director Characteristics

This panel summarizes the characteristics of the directors of our sample firms from 1999 to 2009. The unit of observation is the director-firm-year. "Director with foreign experience" is a dummy equal to one if a director has either foreign education or foreign work experience, and zero otherwise. "Director with foreign work experience" is a dummy equal to one if a director has foreign work experience, and zero otherwise. "Director with foreign education" is a dummy equal to one if a director has foreign education, and zero otherwise. "Foreign visiting scholar/training/postdoc" is a dummy variable equal to one if a director was a visiting scholar, post-doc or did a short-term training program, and zero otherwise. "Foreign bachelor degree" is a dummy variable equal to one if a director holds a bachelor degree from a foreign country, and zero otherwise. "Foreign master degree" is a dummy variable equal to one if a director holds a bachelor degree from a foreign country, and zero otherwise. "Foreign doctoral degree" is a dummy variable equal to one if a director holds a doctoral degree from a foreign country, and zero otherwise. "Age" is the difference between the current year and the birth year of the director. "Female director" is a dummy variable equal to one if the director is female, and zero if male. "Tenure" is one plus the difference between the current year and the year when the individual joined the board of a given firm. "Director employed in the firm" is a dummy variable equal to one if a director also receives a salary as an employee of the firm, and zero otherwise. "Independent director" is a dummy variable equal to one if a director is an independent director, defined according to the China Securities Regulatory Commission as an individual who is not affiliated with the firm except as a director, and does not have a relationship with the company that would interfere with independent judgment. "Foreign director" is a dummy variable equal to one if a director is a foreign national and zero otherwise. "Busy director" is a dummy variable equal to one if a director sits on the boards of two or more publicly traded companies and zero otherwise. "Politically connected director" is a dummy variable equal to one if a director is a current or former government bureaucrat and zero otherwise.

	Mean	Median	Std. Dev.	# of obs.
Director with foreign experience	0.081	0	0.272	138,092
Director with foreign work experience	0.032	0	0.176	138,092
Director with foreign education	0.063	0	0.244	138,092
Foreign visiting scholar/training/postdoc	0.029	0	0.168	138,092
Foreign bachelor degree	0.006	0	0.08	138,092
Foreign master degree	0.017	0	0.129	138,092
Foreign doctoral degree	0.011	0	0.104	138,092
Age	48.203	47	8.893	138,073
Female director	0.097	0	0.296	138,092
Tenure	2.005	2	1.051	138,092
Director employed in the firm	0.391	0	0.488	137,971
Independent director	0.272	0	0.445	138,092
Foreign director	0.004	0	0.062	138,092
Busy director	0.161	0	0.367	138,092
Politically connected director	0.196	0	0.397	138.092

## Table 2 Continued.Panel B: Firm Characteristics

This panel reports the summary statistics for the sample firms between 1999 and 2009. The unit of observation is the firm-year. ROE is the firm's return on equity led by one year. All other variable definitions are in the Appendix.

	Mean	Median	Std. Dev.	# of obs.
MTB	2.401	1.958	1.44	14,068
TFP	0.001	-0.007	0.266	13,144
ROE	0.047	0.065	0.193	13,743
Foreign Experience	0.081	0	0.111	14,331
Number of Directors with Foreign Experience	0.768	0	1.069	14,331
Directors with Foreign Experience Dummy	0.462	0	0.499	14,331
Board Size	9.433	9	2.112	14,331
State	0.696	1	0.46	14,331
Foreign Ownership	0.023	0	0.074	14,331
Block	0.403	0.386	0.167	14,331
Assets (RMB 100 millions)	32.71	14.53	61.96	14,331
Leverage	0.504	0.488	0.25	14,331
ROA	0.028	0.037	0.085	14,331
Average Tenure	1.986	2	0.835	14,331
Average Age	48.096	48.111	3.986	14,331
Busy Directors (Fraction)	0.160	0.125	0.143	14,331
Employed Directors (Fraction)	0.397	0.375	0.221	14,331
Female Directors (Fraction)	0.098	0.091	0.103	14,331
Foreign Directors (Fraction)	0.002	0	0.016	14,331
Board Political Connection	0.194	0.167	0.165	14,331
Top Chinese University (Fraction)	0.026	0	0.059	14,331
Foreign M&A	0.056	0	0.231	4,182
Foreign Private Placement	0.134	0	0.341	350
Foreign Sales	0.128	0.002	0.218	5,259
Earnings Management	0.033	0.019	0.04	13,299
CEO Turnover	0.247	0	0.431	9,027

## Table 2 Continued.Panel C: Industry Distribution of Sample Firms

This panel reports the industry distribution of the sample firms. Statistics are based on firmyear observations. The 21 industries are based on the official industry classification of the China Securities Regulatory Commission.

Industry	%	# of obs.
Agriculture	2.55%	366
Mining	1.64%	235
Food	4.27%	612
Apparel	4.47%	641
Furniture	0.25%	36
Printing	2.04%	293
Gas and Chemistry	11.22%	1,608
Electronic	3.64%	522
Metal	9.33%	1,337
Machinery	15.69%	2,249
Pharmaceutical Products	6.26%	897
Other Manufacturing	1.47%	211
Energy Supply	4.11%	589
Construction	1.97%	282
Transportation	4.12%	591
Information Technology	5.87%	841
Retail & Wholesale	7.06%	1,012
Real Estate	4.05%	580
Other Service Supply	3.17%	454
Entertainment	0.87%	124
Other	5.94%	851

## Table 3Firm Characteristics and Directors with Foreign Experience

We report univariate tests comparing the mean characteristics of firms that eventually hire directors with foreign experience prior to the hiring of any directors with foreign experience to those of firms without directors with foreign experience. The unit of observation is the firm-year and the sample goes from 1999 to 2009. We subtract from MTB, TFP, and ROE their industry-year median as we do in the multivariate analysis. \*\*\*, \*\*, and \* indicate significance at the 1%, 5%, and 10% levels, respectively.

Firm Characteristics	Firms without directors foreign experience		Firms <i>eventi</i> directors experie	Firms <i>eventually</i> with directors foreign experience	
	# of obs.	Mean	# of obs.	Mean	-
MTB	3,442	0.207	2,374	0.221	-0.446
TFP	3,216	-0.013	2,227	0.014	-3.765***
ROE	3,364	-0.028	2,333	-0.029	0.178
Board Size	3,504	9.12	2,398	9.335	-3.695***
Foreign Ownership	3,504	0.005	2,398	0.005	-0.254
Block	3,504	0.414	2,398	0.461	-10.426***
Assets (RMB 100 millions)	3,504	24.732	2,398	19.879	5.726***
Leverage	3,504	0.5	2,398	0.466	5.638***
State	3,504	0.723	2,398	0.816	-8.424***
Foreign M&A	981	0.035	561	0.034	0.082
Foreign Private Placement	88	8%	18	11.10%	-0.387
Foreign Sales	1,266	0.103	381	0.142	-3.336***
Earnings Management	3,188	0.031	2,156	0.045	-11.157***
CEO Turnover	2,240	22.10%	1,618	31%	-6.099***

### Table 4Policy Changes and the Board of Directors

This table relates the proportion of directors with foreign experience ("Foreign Experience") to firm characteristics and the provincial policies. "Provincial Policy" is a dummy variable that takes a value of one if the firm is headquartered in a given province in the years following the adoption of a policy to encourage the return of highly skilled emigrants, and zero otherwise. All other variables are defined in the Appendix. T-statistics, computed with robust standard errors clustered at the year level, are reported in parentheses. All models include a constant and different sets of fixed effects as indicated in the table, but the coefficients are not reported. Columns 3 and 4 also include province-specific linear trends. \*\*\*, \*\*, and \* indicate significance at the 1%, 5%, and 10% levels, respectively.

	(1)	(2)	(3)	(4)
Foreign Ownership	0.467***	0.543***	0.467***	0.399***
	(26.44)	(23.65)	(26.30)	(8.19)
ROA	0.028**	-0.005	0.029**	0.029**
	(2.26)	(-0.42)	(2.27)	(2.26)
Block	-0.016**	-0.027***	-0.015**	-0.013**
	(-2.86)	(-3.61)	(-2.80)	(-2.55)
State	-0.011***	-0.015***	-0.011***	-0.013***
	(-4.09)	(-4.93)	(-4.07)	(-3.65)
Size	0.008***	0.005***	0.008***	0.008***
	(7.28)	(3.18)	(7.19)	(7.26)
Leverage	0.002	-0.003	0.002	0.002
	(0.48)	(-0.74)	(0.42)	(0.47)
Provincial Policy			0.012**	0.009**
			(3.05)	(2.29)
Provincial Policy × State				0.004
				(1.43)
Provincial Policy × Foreign Ownership				0.102*
				(2.08)
Provincial Policy × Block				-0.004
				(-0.63)
Province-Specific Trend	Yes	Yes	Yes	Yes
Industry FE	Yes	No	Yes	Yes
Firm FE	No	Yes	No	No
F-test of excluded instruments			9.33**	13.72***
# of obs.	14,331	14,331	14,331	14,331
R-squared	0.19	0.68	0.19	0.19

### Table 5Directors with Foreign Experience and Firm Value

This table relates the firm's market to book ratio (MTB) to the proportion of directors with foreign experience. In columns 1 to 3, we present ordinary least squares estimates. In columns 4 to 6, we present instrumental variable estimates. In columns 1 to 5, the dependent variable is the firm's MTB from which we subtract the industry-year median. The dependent variable in column 6 is the firm's MTB from which we subtract the industry-year median and the province-year median. The instrumental variables in columns 4 to 6 includes the "Provincial Policy", a dummy variable that takes a value of one in years following the implementation of the policy in each province and interaction variables between the policy dummy and firm characteristics "State", "Foreign Ownership", and "Block" in 1999 (the beginning of the sample period). If a firm enters our sample later than 1999, firm characteristics computed in the year of the firm's entry in the sample are used to construct the interaction terms. All the variables are defined in the Appendix. T-statistics computed with robust standard errors clustered at firm level are reported in parentheses. All models include a constant, but the coefficients are not reported. \*\*\*, \*\*, and \* indicate significance at the 1%, 5%, and 10% levels, respectively.

	OLS				IV		
	(1)	(2)	(3)	(4)	(5)	(6)	
						Within- Province Test	
Foreign Experience	0.943***	0.851***	0.463**	5.554***	8.838***	17.759***	
	(6.94)	(5.79)	(2.57)	(3.91)	(4.50)	(5.55)	
Foreign Ownership		0.433*	0.729	-1.897***	-3.438***	-8.132***	
		(1.74)	(1.41)	(-2.64)	(-3.60)	(-4.76)	
Block		0.529***	-0.483***	0.685***	0.701***	0.989***	
		(5.48)	(-2.83)	(5.64)	(4.74)	(3.82)	
State		-0.151***	-0.118**	-0.061	-0.028	0.076	
		(-4.12)	(-2.29)	(-1.17)	(-0.43)	(0.70)	
Size	-0.517***	-0.524***	-0.635***	-0.578***	-0.614***	-0.657***	
	(-25.69)	(-25.40)	(-17.81)	(-18.97)	(-16.57)	(-10.71)	
Leverage	0.123	0.172*	0.510***	0.194*	0.181	0.473***	
	(1.21)	(1.66)	(4.67)	(1.74)	(1.49)	(2.82)	
Firm FE	No	No	Yes	No	No	No	
Province FE	No	No	No	No	Yes	No	
# of obs.	14,068	14,068	14,068	14,068	14,068	14,068	
R-squared	0.20	0.21	0.11	0.21	0.23	0.16	

### Table 6 Directors with Foreign Experience and Total Factor Productivity

This table relates the firm's total factor productivity (TFP) to the presence of directors with foreign experience. In columns 1 to 3, we present ordinary least squares estimates. In columns 4 to 6, we present instrumental variable estimates. In columns 1 to 5, the dependent variable is the firm's TFP from which we subtract the industry-year median. The dependent variable in column (6) is the firm's TFP from which we subtract the industry-year median and the province-year median. The instrumental variables in columns 4 to 6 includes the "Provincial Policy", a dummy variable that takes a value of one in years following the implementation of the policy in each province and interaction variables between the policy dummy and firm characteristics "State", "Foreign Ownership", and "Block" in 1999 (the beginning of the sample period). If a firm enters our sample later than 1999, firm characteristics computed in the year of the firm's entry in the sample are used to construct the interaction terms. All the variables are defined in the Appendix. T-statistics computed with robust standard errors clustered at firm level are reported in parentheses. All models include a constant, but the coefficients are not reported. \*\*\*, \*\*, and \* indicate significance at the 1%, 5%, and 10% levels, respectively.

	OLS				IV		
	(1)	(2)	(3)	(4)	(5)	(6)	
						Within- Province Test	
Foreign Experience	0.114***	0.139***	0.082**	0.981***	0.759**	0.838**	
	(2.89)	(3.28)	(2.02)	(2.86)	(2.26)	(2.53)	
Foreign Ownership		-0.032	-0.053	-0.462**	-0.364**	-0.421**	
		(-0.53)	(-0.50)	(-2.53)	(-2.12)	(-2.40)	
Block		0.099***	0.039	0.131***	0.107***	0.113***	
		(3.48)	(1.02)	(4.09)	(3.49)	(3.63)	
State		0.005	-0.008	0.023*	0.019	0.021	
		(0.50)	(-0.61)	(1.76)	(1.50)	(1.63)	
Size	-0.001	-0.004	-0.036***	-0.014**	-0.011*	-0.011*	
	(-0.19)	(-0.94)	(-5.32)	(-2.12)	(-1.81)	(-1.86)	
Leverage	-0.123***	-0.111***	-0.065***	-0.109***	-0.111***	-0.110***	
	(-6.51)	(-5.82)	(-2.93)	(-5.12)	(-5.48)	(-5.41)	
Firm FE	No	No	Yes	No	No	No	
Province FE	No	No	No	No	Yes	No	
# of obs.	13,144	13,144	13,144	13,144	13,144	13,144	
R-squared	0.01	0.02	0.01	0.02	0.03	0.02	

### Table 7Directors with Foreign Experience and Profitability

This table relates the firm's profitability to the presence of directors with foreign experience. In columns 1 to 3, we present ordinary least squares estimates. In columns 4 to 6, we present instrumental variable estimates. In columns 1 to 5, the dependent variable is the firm's ROE at t + 1 from which we subtract the industry-year median. The dependent variable in columns 6 is the firm's ROE at t + 1 from which we subtract the industry-year median. The dependent variable in columns 6 is the firm's ROE at t + 1 from which we subtract the industry-year median and the province-year median. The instrumental variables in columns 4 to 6 includes the "Provincial Policy", a dummy variable that takes a value of one in years following the implementation of the policy in each province and interaction variables between the policy dummy and firm characteristics "State", "Foreign Ownership", and "Block" in 1999 (the beginning of the sample period). If a firm enters our sample later than 1999, firm characteristics computed in the year of the firm's entry in the sample are used to construct the interaction terms. All the variables are defined in the Appendix. T-statistics computed with robust standard errors clustered at firm level are reported in parentheses. All models include a constant, but the coefficients are not reported. \*\*\*, \*\*, and \* indicate significance at the 1%, 5%, and 10% levels, respectively.

	OLS				IV					
	(1)	(2)	(3)	(4)	(5)	(6)				
						Within- Province Test				
Foreign Experience	0.054***	0.063***	0.022	0.582***	0.477**	0.392**				
	(2.90)	(3.16)	(0.77)	(2.88)	(2.10)	(2.06)				
Foreign Ownership		-0.037	0.060	-0.293***	-0.244**	-0.216**				
		(-1.18)	(0.73)	(-2.59)	(-2.07)	(-2.04)				
Block		0.073***	0.132***	0.091***	0.086***	0.088***				
		(5.82)	(5.08)	(5.69)	(5.56)	(5.87)				
State		-0.021***	-0.032***	-0.011*	-0.009	-0.011*				
		(-4.27)	(-3.28)	(-1.74)	(-1.39)	(-1.74)				
Size	0.017***	0.017***	-0.039***	0.010***	0.012***	0.010***				
	(7.85)	(7.44)	(-8.02)	(2.92)	(3.35)	(3.15)				
Leverage	-0.039***	-0.033**	0.180***	-0.028*	-0.029*	-0.029*				
	(-2.65)	(-2.21)	(7.06)	(-1.74)	(-1.84)	(-1.91)				
Firm FE	No	No	Yes	No	No	No				
Province FE	No	No	No	No	Yes	No				
# of obs.	13,743	13,743	13,743	13,743	13,743	13,743				
R-squared	0.01	0.02	0.02	0.02	0.02	0.01				

### Table 8Corporate Policies

This table relates corporate policies to the proportion of directors with foreign experience. The dependent variable in columns 1 to 3 of Panel A is "Foreign M&A", a dummy variable equal to one if at least one of the mergers and acquisitions a sample firm announced in a given year involves a foreign firm, and zero otherwise. In columns 4 and 5, the dependent variable is "Foreign Private Placement", a dummy variable equal to one if at least one of the firm's private placements in a given year is targeted to foreign investors and zero if none of these private placements is targeted to foreign investors. In columns 6 to 7 of Panel A, the dependent variable is "Foreign Sales" from which we subtract the industry-year median. The dependent variable in columns 1 and 2 of Panel B is a proxy increasing in the extent of earnings management in the spirit of Stubben (2010). The dependent variable in column 3 of Panel B is a dummy variable that takes value 1 if there is an event of CEO turnover and value zero otherwise. In columns 1, 3, 4, and 6 of Panel A and column 1 and 3 of Panel B, estimates are obtained by ordinary least squares; in the rest of the columns we report instrumental variable estimates. The instrumental variables in columns 4 to 6 includes the "Provincial Policy", a dummy variable that takes a value of one in years following the implementation of the policy in each province and interaction variables between the policy dummy and firm characteristics "State", "Foreign Ownership", and "Block" in 1999 (the beginning of the sample period). If a firm enters our sample later than 1999, firm characteristics computed in the year of the firm's entry in the sample are used to construct the interaction terms. Variable definitions are in the Appendix. T-statistics computed with robust standard errors clustered at firm level are reported in parentheses. All models include a constant, but the coefficients are not reported. \*\*\*, \*\*, and \* indicate significance at the 1%, 5%, and 10% levels, respectively.

	Foreign M&A		Foreign Priva	te Placement	Foreign Sales		
	OLS	IV	OLS	OLS	IV	OLS	IV
	(1)	(2)	(3)	(4)	(5)	(6)	(7)
Foreign Experience	0.203***	0.844***	0.127***	0.320*	0.417	0.260***	0.703***
	(4.18)	(3.99)	(2.88)	(1.74)	(0.64)	(4.23)	(3.20)
Foreign Experience × Same Country M&A			2.986***				
			(9.25)				
Dummy for B/H-share				-0.095	-0.098		
				(-1.60)	(-1.51)		
State	-0.016*	-0.006	-0.014	-0.048	-0.043	-0.000	0.008
	(-1.88)	(-0.64)	(-1.64)	(-0.99)	(-0.88)	(-0.04)	(0.65)
Size	0.014**	0.006	0.012**	0.015	0.012	-0.009*	-0.015***
	(2.55)	(1.08)	(2.45)	(0.81)	(0.53)	(-1.78)	(-2.68)
Leverage	-0.018	-0.020	-0.019	-0.126	-0.121	-0.010	-0.000
	(-0.88)	(-0.91)	(-1.00)	(-1.30)	(-1.08)	(-0.38)	(-0.01)
MTB	0.002	0.005	0.002	-0.012	-0.010	-0.004	-0.005
	(0.65)	(1.57)	(0.50)	(-0.92)	(-0.77)	(-1.18)	(-1.46)
ROA	-0.031	-0.045	-0.033	0.422	0.501	0.091	0.082
	(-0.53)	(-0.69)	(-0.61)	(1.36)	(1.52)	(1.36)	(1.20)
Year FE	Yes	No	Yes	Yes	No	No	No
Industry FE	Yes	Yes	Yes	Yes	Yes	No	No
# of obs.	4,118	4,118	4,118	344	344	5,170	5,170
R-squared	0.03	0.04	0.11	0.13	0.12	0.02	0.03

# Table 8 continued.Panel A: Internationalization

	Earnings N	lanagement	CEO Turnover
	OLS	IV	OLS
	(1)	(2)	(3)
Foreign Experience	-0.013**	-0.484***	0.029
	(-2.05)	(-6.19)	(0.63)
Foreign Experience × ROA			-0.881*
			(-1.81)
ROA	0.007	0.001	-0.571***
	(0.87)	(0.06)	(-6.56)
State	0.009***	-0.003	0.012
	(4.43)	(-0.86)	(1.17)
Size	-0.009***	-0.002	-0.020***
	(-9.35)	(-1.06)	(-4.06)
Leverage	-0.011***	-0.007	0.019
	(-3.02)	(-1.45)	(0.80)
CEO Age			0.006***
			(8.18)
CEO Tenure			-0.013***
			(-6.15)
Firm FE	Yes	Yes	No
# of obs.	13,299	13,243	9,027
R-squared	0.02	0.03	0.03

# Table 8 continued.Panel B: Corporate Governance

### Table 9 Direct and Indirect Effects of the Provincial Policies

The dependent variable is the firm's market to book ratio (MTB) from which we subtract the province-year median in column (1), the firm's total factor productivity (TFP) from which we subtract the province-year median in column (2), and the ROE at year t+1 from which we subtract the province-year median in column (3). Variable definitions are in the Appendix. T-statistics computed with robust standard errors clustered at firm level are reported in parentheses. All models include a constant, but the coefficients are not reported. \*\*\*, \*\*, and \* indicate significance at the 1%, 5%, and 10% levels, respectively.

	MTB	TFP	ROE (t+1)
	(1)	(2)	(3)
Provincial Policy × Foreign Experience	0.490***	0.120***	0.054**
	(3.08)	(2.72)	(2.51)
Provincial Policy	-0.036	-0.011	-0.004
	(-0.95)	(-1.08)	(-0.60)
Foreign Ownership	0.750***	-0.036	-0.031
	(3.09)	(-0.61)	(-0.98)
Block	0.657***	0.107***	0.088***
	(6.44)	(3.58)	(6.39)
State	-0.135***	0.012	-0.016***
	(-3.63)	(1.24)	(-3.11)
Size	-0.597***	-0.007	0.014***
	(-28.58)	(-1.40)	(5.48)
Leverage	0.131	-0.128***	-0.040**
	(1.24)	(-6.68)	(-2.53)
Year FE	Yes	Yes	Yes
Industry FE	Yes	Yes	Yes
# of obs.	14,068	13,144	13,743
R-squared	0.26	0.02	0.03

#### Table 10

#### **Proportion of Directors with Foreign Experience and Other Board Characteristics**

This table relates the proportion of directors with foreign experience to firm performance controlling for additional board characteristics. The dependent variable is the market to book ratio (MTB) from which we subtract the industry-year median in column 1, the total factor productivity (TFP) from which we subtract the industry-year median in column 2, and the ROE at year t + 1 from which we subtract the industry-year median in column 3. The additional controls are: "Average Tenure", defined as the average tenure of the firm's directors; "Employed Directors (Fraction)", defined as the proportion of directors that receive a salary from the firm; "Female Directors (Fraction)", defined as the proportion of female directors in the firm's board; "Average Age", defined as the average age of the directors; "Foreign Directors (Fraction)", defined as the proportion of foreign directors in the firm's board; "Board Political Connections", defined as the proportion of current or former government bureaucrats as in Fan et al. (2007) and Calomiris et al. (2010); and "Top Chinese University (Fraction)", defined as the fraction of directors that graduated from and/or worked for Peking University and/or Tsinghua University, the two top universities in China. The rest of variables are defined in the Appendix. For brevity we report only the instrumental variable estimates. The instrumental variables include "Provincial Policy", a dummy variable that takes a value of one in the years following the implementation of the policy in each province and zero otherwise, and the interactions between this dummy with firm characteristics "State", "Foreign Ownership", and "Block" in 1999 (the beginning of the sample period). If a firm enters our sample later than 1999, firm characteristics computed in the year of the firm's entry in the sample are used to construct the interaction terms. T-statistics computed with robust standard errors clustered at firm level are reported in parentheses. All models include a constant, but the coefficients are not reported. \*\*\*, \*\*, and \* indicate significance at the 1%, 5%, and 10% levels, respectively.

	MTB	TFP	ROE (t+1)
	(1)	(2)	(3)
Foreign Experience	5.037**	1.054**	0.708**
	(2.50)	(2.15)	(2.38)
Foreign Ownership	-1.184	-0.430**	-0.300**
	(-1.53)	(-2.13)	(-2.31)
Block	0.637***	0.118***	0.086***
	(5.63)	(3.84)	(5.44)
State	-0.120**	0.015	-0.011
	(-2.13)	(1.04)	(-1.43)
Size	-0.574***	-0.018***	0.010***
	(-20.81)	(-3.06)	(3.08)
Leverage	0.183*	-0.104***	-0.026
	(1.66)	(-4.98)	(-1.59)
Average Tenure	0.025*	0.002	0.006***
	(1.83)	(0.46)	(2.97)
Employed Directors (Fraction)	-0.225**	0.050*	0.028*
	(-2.05)	(1.77)	(1.66)
Board Size	0.010	0.006***	-0.000
	(1.32)	(3.02)	(-0.17)
Female Directors (Fraction)	-0.078	-0.014	0.046*
	(-0.43)	(-0.33)	(1.73)
Average Age	0.007	0.004***	0.001
	(1.44)	(3.54)	(1.44)
Foreign Directors (Fraction)	-6.093**	-1.167*	-0.676*
	(-2.05)	(-1.67)	(-1.70)
Board Political Connection	0.170	-0.027	0.008
	(1.61)	(-0.84)	(0.49)
Top Chinese University (Fraction)	-1.278	-0.311	-0.253**
	(-1.59)	(-1.56)	(-2.23)
Busy Directors (Fraction)	-0.146	0.003	-0.003
	(-0.65)	(0.06)	(-0.10)
# of obs.	14,068	13,144	13,743
R-squared	0.22	0.03	0.02

#### Table 10 continued.

#### Table 11

#### **Controlling for Firm Unobserved Heterogeneity**

This table reports instrumental variable estimates for a sample of firms with and without foreign experienced matched on year, industry, size and leverage using a nearest neighbor estimator and for the full sample; in the latter, we control for firm fixed effects and the lagged performance. The dependent variable is the market to book ratio (MTB) (column 1 and 4), total factor productivity (TFP) (column 2 and 5), and ROE at year t + 1 (column 3 and 6), from which we subtract the industry-year median, respectively. In columns 5 to 6, we match firms with directors with foreign experience to firms with firms that on the basis of their size, leverage industry and year have similar probability (probability) of having directors with foreign experience. We include in the control sample only firms with similar propensity score. All models are estimated by instrumental variables The instrumental variables include "Provincial Policy", a dummy variable that takes a value of one in the years following the implementation of the policy in each province and zero otherwise, and the interactions between this dummy with firm characteristics "State", "Foreign Ownership", and "Block" in 1999 (the beginning of the sample period). If a firm enters our sample later than 1999, firm characteristics computed in the year of the firm's entry in the sample are used to construct the interaction terms. Variable definitions are in the Appendix. T-statistics computed with robust standard errors clustered at firm level are reported in parentheses. All models include a constant, but the coefficients are not reported. \*\*\*, \*\*, and \* indicate significance at the 1%, 5%, and 10% levels, respectively.

	Matched Sample				Full Sample				
	MTB	TFP	ROE (t+1)	-	MTB	TFP	ROE (t+1)		
	(1)	(2)	(3)		(4)	(5)	(6)		
Foreign Experience	3.810**	0.895*	0.597**		14.983***	1.140**	0.560**		
	(2.13)	(1.89)	(2.04)		(5.34)	(2.25)	(2.04)		
Foreign Ownership	-1.110	-0.389*	-0.282*		-7.561***	-0.609**	-0.232		
	(-1.35)	(-1.72)	(-1.91)		(-3.88)	(-1.98)	(-1.36)		
Block	0.613***	0.107***	0.088***		0.021	0.044	0.155***		
	(5.04)	(3.05)	(4.79)		(0.07)	(1.01)	(5.28)		
State	-0.096*	0.018	-0.012		0.118	0.012	-0.014		
	(-1.75)	(1.18)	(-1.53)		(1.19)	(0.72)	(-1.22)		
Size	-0.498***	-0.008	0.015***		-0.675***	-0.045***	-0.036***		
	(-18.52)	(-1.30)	(4.32)		(-11.83)	(-5.20)	(-6.10)		
Leverage	0.076	-0.126***	-0.045***		0.530***	-0.056**	0.093***		
	(0.65)	(-5.04)	(-2.71)		(3.91)	(-2.24)	(4.04)		
Lag(MTB)					0.187***				
					(12.56)				
Lag(TFP)						0.213***			
						(11.92)			
ROE							0.138***		
							(6.80)		
Firm FE	No	No	No		Yes	Yes	Yes		
# of obs.	10,108	9,453	9,894		13,013	11,208	13,505		
R-squared	0.19	0.02	0.02		0.27	0.24	0.01		

### Table 12The Impact of Quality of Foreign Education

This table relates alternative measures of the directors' foreign experience to firm performance. "Foreign Experience (Weighted by Foreign Degree)" is computed as number of directors with a foreign academic degree scaled by total number of directors of the firm. We present an *F*-test for the difference of the coefficients of Foreign Experience and Foreign Experience (Weighted by Foreign Degree). The dependent variable is the market to book ratio (MTB) from which we subtract the industry-year median in columns 1 and 2, the total factor productivity (TFP) from which we subtract the industry-year median in column 3 and 4, and the ROE at year t + 1 from which we subtract the industry-year median in columns 5 and 6. Estimates are obtained by ordinary least squares. Variable definitions are in the Appendix. T-statistics computed with robust standard errors clustered at firm level are reported in parentheses. All models include a constant, but the coefficients are not reported. \*\*\*, \*\*, and \* indicate significance at the 1%, 5%, and 10% levels, respectively.

	MTB		TFP		ROE (t+1)	
	(1)	(2)	(3)	(4)	(5)	(6)
Foreign Experience	0.463**		0.082**		0.022	
	(2.57)		(2.02)		(0.77)	
Foreign Experience (Weighted by Foreign Degree)		0.536*		0.157**		0.067*
		(1.91)		(2.45)		(1.77)
Foreign Ownership	0.729	0.794	-0.053	-0.061	0.060	0.049
	(1.41)	(1.56)	(-0.50)	(-0.60)	(0.73)	(0.60)
Block	-0.483***	-0.494***	0.039	0.039	0.132***	0.133***
	(-2.83)	(-2.89)	(1.02)	(1.03)	(5.08)	(5.12)
State	-0.118**	-0.120**	-0.008	-0.008	-0.032***	-0.032***
	(-2.29)	(-2.35)	(-0.61)	(-0.58)	(-3.28)	(-3.22)
Size	-0.635***	-0.632***	-0.036***	-0.036***	-0.039***	-0.040***
	(-17.81)	(-17.73)	(-5.32)	(-5.31)	(-8.02)	(-8.03)
Leverage	0.510***	0.514***	-0.065***	-0.065***	0.180***	0.180***
	(4.67)	(4.70)	(-2.93)	(-2.91)	(7.06)	(7.07)
Firm FE	Yes	Yes	Yes	Yes	Yes	Yes
F-test. H <sub>0</sub> : Foreign Experience = Foreign Experience (Weighted by Foreign Degree)	1.	22	4.0	6**	2.	53
# of obs.	14,068	14,068	13,144	13,144	13,743	13,743
R-squared	0.11	0.11	0.01	0.01	0.02	0.02

Table 12 continued.